Check for updates

OPEN ACCESS

EDITED BY Dmytro Chumachenko, National Aerospace University – Kharkiv Aviation Institute, Ukraine

REVIEWED BY

Zaijun Li, Yangzhou University, China Zhongyin Wei, Southwest University, China

*CORRESPONDENCE Heng Chen 🖂 chenhengpeerless@163.com

RECEIVED 08 July 2024 ACCEPTED 17 February 2025 PUBLISHED 06 March 2025

CITATION

Chen H, Lin X, Guo Y and Qi X (2025) Heterogeneity and spatiotemporal evolution characteristics of regional emergency logistics response capacity: a case of China. *Front. Public Health* 13:1461354. doi: 10.3389/fpubh.2025.1461354

COPYRIGHT

© 2025 Chen, Lin, Guo and Qi. This is an open-access article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.

Heterogeneity and spatiotemporal evolution characteristics of regional emergency logistics response capacity: a case of China

Heng Chen[®]*, Xianglong Lin, Yuan Guo and Xianchao Qi

School of Management, Xi'an Polytechnic University, Xi'an, China

Public emergencies have surged worldwide, highlighting the critical role of emergency logistics in disaster relief. This study evaluates the heterogeneity and spatiotemporal evolution characteristics of regional emergency logistics response capacity in China using the entropy-weighted TOPSIS method, Dagum Gini coefficient, kernel density estimation, and Markov chain analysis. The emergency logistics response capacities of 30 provinces and four major economic regions (Eastern, Central, Western, Northeastern) were assessed from 2012 to 2021. Key findings reveal: (1) Provincial emergency logistics response capacities improved significantly nationwide, yet regional imbalances remain pronounced. (2) Regional heterogeneity expanded, driven primarily by inter-regional disparities. (3) Temporal analysis shows steady growth across all economic regions without polarization, with the Eastern region achieving the highest mean capacity and growth rate. (4) Spatial evolution demonstrates continuity, as Markov chain analysis reveals gradual transitions between adjacent capacity levels but limited leapfrog development. Spatial factors exert dual effects: proximity to high-capacity regions facilitates upgrades for moderate-level areas, yet suppresses low capacity regions. These findings emphasize persistent structural gaps in infrastructure, resource allocation, and governance.

KEYWORDS

emergency logistics response capacity, regional heterogeneity, spatiotemporal evolution characteristics, entropy-weighted TOPSIS method, Markov chain

1 Introduction

In recent years, there has been an observable trend toward an increased intensity and frequency of natural disasters and other sudden public emergencies, indicating a new normal (1). China, being one of the countries most severely affected by natural disasters, is characterized by a diverse array of disaster types, a wide geographic distribution, frequent occurrences, and substantial losses (2). The "2021 Global Natural Disaster Assessment Report" highlights that China was struck by 21 significant natural disasters in 2021 alone, affecting 107 million people and leading to 765 fatalities and 102 missing persons, with direct economic losses amounting to an alarming 334.02 billion yuan (46.15 billion dollar). These public emergencies not only result in severe human casualties and financial losses but also pose formidable challenges to the disaster management and emergency response systems (3). Within this context, emergency logistics emerges as a pivotal element, providing essential support for the supply demands and personnel deployment during such crises (4). The effectiveness of emergency logistics is crucial for mitigating disaster impacts and ensuring the prompt delivery of necessary goods and services to the affected regions (5). Accordingly, this

aspect of emergency management has attracted considerable attention from academic circles, governmental bodies, and the public at large.

Numerous scholars have investigated the assessment of emergency logistics response capacity employing methodologies such as the Analytic Hierarchy Process (AHP) and Fuzzy Comprehensive Evaluation (FCE) (6–9). However, these existing methodologies exhibit limitations due to the subjectivity and randomness in selecting indicator weights, compromising the scientific validity and reliability of the evaluations. Despite these challenges, some researchers have applied case study methods to delve into emergency logistics in specific contexts (10-12), focusing predominantly on particular disasters or locales from a micro perspective, thus limiting broader regional evaluations of emergency logistics.

Additionally, the comprehensive nature of emergency logistics response capacity (Hereinafter referred to as ELRC) remains underexplored in the literature, further complicated by significant regional disparities in China's logistics development (13), which have led to evident regional heterogeneity in ELRC. Recent initiatives in China have included the introduction of coordinated regional logistics development plans, which have significantly influenced the landscape of regional logistics development of regional logistics development patterns. However, existing research has yet to adequately address the dynamic characteristics of ELRC, thus leaving a gap in the literature regarding its evolving nature.

The objectives of this study are threefold: First, it aims to develop a comprehensive indicator system to evaluate ELRC, enabling the scientific and objective measurement across different regions in China. Second, the study analyzes regional heterogeneities in China's ELRC. Third, it investigates the spatiotemporal evolution of these capacities. This research offers a novel perspective and theoretical foundation, enhancing the understanding of China's regional ELRC. The findings are intended to provide valuable insights for crafting more scientifically robust and rational emergency logistics policies. This study adopts a regional perspective, broadening the scope of existing literature by providing an in-depth analysis of the factors constituting ELRC and establishing an integrative index system. The entropy-weighted TOPSIS method is employed for measurement, enhancing the rigor and credibility of the results. The use of the Dagum Gini coefficient for heterogeneity analysis introduces a novel approach for understanding inter-regional disparities. Furthermore, the spatiotemporal evolution analysis enhances the understanding of the changing patterns and characteristics of emergency logistics. The paper is organized as follows: Section 2 provides a literature review. Section 3 describes the research methodology and data sources. Section 4 outlines the components and measurement results of ELRC. Section 5 discusses the heterogeneity and origins of regional differences in ELRC within China. Section 6 examines the spatiotemporal evolution of these capacities. Section 7 discusses the study's novelty, significance, limitations, and directions for future research. Section 8 concludes with the study's implications and recommendations for policy.

2 Literature review

In the field of emergency logistics, contemporary research has primarily focused on three distinct areas: the concept and connotation, the various patterns and models, and the methods of evaluation.

2.1 Concept and connotation of emergency logistics

Emergency logistics, internationally recognized as a wellestablished discipline, is defined as the urgent logistical activities designed to ensure efficient transportation and allocation of relief materials, equipment, and personnel in response to unforeseen events such as public health emergencies and natural disasters. Its core objective lies in delivering critical resources to disasterstricken areas within the shortest possible timeframe to minimize losses and preserve lives (14). Key characteristics of this field include its unpredictability and exceptionality (15, 16), stochastic demand and selectivity (17, 18), uneven flow distribution (19, 20), time-critical constraints, and the imperative to prioritize social welfare (21). Due to the inherent unpredictability of such events, emergency logistics frequently encounters challenges stemming from the scarcity of historical planning data (22). Nevertheless, it has evolved into an indispensable component of disaster mitigation strategies, playing a pivotal role in managing crises effectively and delivering humanitarian aid (23).

Distinct from conventional social logistics, which aims to sustain societal operations and enhance quality of life (24), emergency logistics prioritizes time-efficiency optimization and loss minimization (25). Despite this divergence, scholars have identified shared foundational components between the two systems, including carriers, flow direction, flow, and flow rate (14). Consequently, evaluations of ELRC must be grounded in these intrinsic characteristics. Given the resource scarcity inherent to emergency logistics and its frequent deviation from full decision-maker control (26), a robust material foundation—encompassing infrastructure, transportation assets, and skilled personnel—is critical to ensure timely support for affected populations through coordinated material, human, informational, and service flows (27, 28).

Functionally, emergency logistics encompasses three primary tasks: (1) delivering relief supplies, (2) evacuating and relocating affected populations, and (3) transporting casualties to medical facilities to ensure safety in disaster zones (29). This necessitates enhanced resource allocation capabilities, operationalized through metrics such as freight volume and turnover rates, to guarantee both scale and efficiency in material distribution. Simultaneously, logistics velocity and operational efficacy significantly influence overall responsiveness. First, response capacity correlates closely with roadway hierarchy and transportation network density, which determine flow velocity and directly impact timeliness. Second, the ratio of logistics output to resource input serves as a critical indicator of utilization efficiency, reflecting the operational effectiveness of response systems—a particularly vital consideration in resource-constrained emergency contexts.

It is noteworthy that while transportation metrics such as freight volume and road network density are conventionally employed to gauge industrial development levels, extant research demonstrates their critical role in reflecting system redundancy and response resilience within emergency logistics contexts (27). For instance, Sheu (27) empirically validated through case studies that the spatial density of road freight volume exhibits a significant positive correlation with disaster relief allocation efficiency. Further building on this, Zhang et al. (30) concluded that the proportion of high-grade roadway has a direct impact on supply chain reliability under extreme conditions, and thus the proportion of high-grade roadway can be identified as a key predictor of emergency logistics timeliness. Thus, this study contends that transportation metrics in ELRC assessments transcend mere industrial scale representation, instead deriving their analytical value from context-specific explanatory power in crisis scenarios.

Beyond these foundational components, emergency logistics serving as a pivotal support framework and critical safeguard mechanism for public emergencies—necessitates integration with real-world contextual factors, particularly governmental coordination mechanisms and digitalization initiatives, to address the complexities inherent in contemporary crisis management.

2.2 Patterns and models of emergency logistics

In the domain of emergency logistics, researchers have innovatively developed various efficient logistics and resource allocation patterns by incorporating advanced technologies such as cloud platforms, artificial intelligence, and blockchain (31-33). Concurrently, Banomyong and Sopadang enhanced the emergency logistics response model by applying the Monte Carlo simulation technique (10). Similarly, Liu et al. (34) developed data-driven models for optimizing emergency logistics networks, while Wang et al. (35) and Chen et al. (36) utilized simulation techniques to evaluate and enhance the reliability and design of these networks.

Further expanding the scope of this research, scholars both domestically and internationally have devoted considerable effort to refining models for emergency logistics facility location selection (37–40), transportation route planning for rescue materials (41–44), and integrated location-path planning (45–47). In addition, models focusing on emergency logistics decision planning (47–49) and regional emergency dispatch (50, 51) have also received significant scholarly attention and discussion.

2.3 Evaluation of emergency logistics

In the field of emergency logistics evaluation research, scholars have extensively explored aspects related to reliability, performance, and capability. Initially, Chen et al. (6) introduced the concept of emergency logistics supply chain reliability and evaluated it using the Analytic Hierarchy Process (AHP). Subsequently, Gong et al. (52) conducted a case study to assess the reliability of the emergency logistics system, employing the Fuzzy Entropy Clustering Method for their evaluation. Xu et al. (7) further developed an emergency supply chain reliability evaluation model incorporating assurance mechanisms and information systems, utilizing the AHP and Entropy Weight Method (EWM). They validated the effectiveness and applicability of this model through practical examples. Additionally, Zhang et al. (53) explored the essence of urban emergency logistics system reliability, integrating practical cases and applying the Fuzzy Comprehensive Evaluation (FCE) to assess its reliability.

Regarding performance evaluation, Liu et al. (54) utilized the fuzzy entropy-weighted TOPSIS method with Multi-Granularity Linguistic Assessment¹ (MGLA) information to assess emergency logistics performance in five severely affected areas during the Wenchuan earthquake. Ji and Zhang (55) proposed a performance evaluation method based on the extensible matter-element model, which resolved incompatibility and contradictions among evaluation indicators, ensuring objectivity and scientificity in performance assessments. Zhang et al. (56) developed an evidence-based emergency logistics performance evaluation model using a specific evaluation method, validated through an instance involving the distribution of emergency resources for flood control along the Jingjiang River. Li et al. (57) adopted multiple methods, including the Delphi Method, AHP, Grey System Theory, and Fuzzy Evaluation, to achieve the most rational performance evaluation results at different stages of the emergency logistics system.

In the domain of capability evaluation, Deng et al. (9) constructed an evaluation index system for ELRC, assessing it using the Fuzzy Grey Comprehensive Evaluation method. Zhang et al. (58) established a dynamic evaluation model of emergency logistics capability based on the evaluation index system for COVID-19 and combined it with a BP neural network. Yang et al. (59) adopted a multi-attribute decisionmaking technique grounded in probabilistic linguistic terminology for the evaluation of ELRC. Lin et al. (8) developed an evaluation index system for natural disaster emergency logistics capabilities and used the Analytic Network Process (ANP) and multi-level grey evaluation method for the assessment. Sun et al. (60) applied the Fuzzy-AHP to evaluate the ELRC during the Wenchuan, Qinghai, and Lushan earthquakes. Chen and Ke (61) effectively evaluated emergency capabilities comprehensively for five emergency logistics decisionmaking units using the FCE method based on data envelopment analysis.

Upon systematically summarizing existing research achievements, it is apparent that both domestic and international scholars have explored the concept and connotation of emergency logistics, leading to the establishment of a well-defined theoretical structure and research system. However, the existing relevant research results predominantly focus on qualitative research and have not been further applied to quantitative analysis of ELRC. Regarding research on emergency logistics patterns and model optimization, many studies approach the topic from the mathematical model construction or simulation perspective, focusing mainly on micro-level emergency logistics decision-making and scheduling aspects. Nevertheless, there is a lack of emphasis on identifying regional ELRC. The ultimate objective of both emergency logistics decision-making and scheduling is to enhance ELRC. However, without effectively discerning the current state and spatiotemporal evolution characteristics of ELRC in different regions, both site selection and decision-making in emergency logistics would eventually lack a solid foundation.

Currently, in emergency logistics evaluation research, scholars have discussed various perspectives, such as reliability, performance, and capabilities. However, the existing evaluation methods chosen for study often lean toward subjective approaches, such as the AHP and FCE. Relying on subjective factors to select evaluation indicators may lead to potential inadequacies and a lack of comprehensiveness in the

¹ Multi-Granularity Linguistic Assessment (MGLA) is a method used in decision-making processes that involves evaluating and synthesizing information based on linguistic terms of different granularity levels, allowing for more nuanced and flexible analysis of subjective data.

design of evaluation systems. Consequently, the scientificity and accuracy of evaluation results may be undermined. Furthermore, most studies adopt case study methods, focusing primarily on specific locations or particular natural disasters from a micro-level perspective, lacking exploration of a macro-regional view. Moreover, emergency logistics is a complex and multifaceted system, and ELRC is a comprehensive capability. The existing research frequently fails to reflect all the constituent factors involved fully.

Based on existing domestic and international research, this study conducts several extensions. Firstly, it analyzes the constituent factors contributing to ELRC and develops a comprehensive index system. Then, the index system is utilized with the entropy-weighted TOPSIS method to measure the ELRC of 30 provincial administrative regions and four major economic regions in China from 2012 to 2021. Secondly, the study explores the regional heterogeneity of China's ELRC and its sources using the Dagum Gini coefficient. Thirdly, the study utilizes kernel density estimation and Markov chain analysis methods to examine the spatiotemporal evolution process and characteristics of China's regional ELRC.

3 Research methods and data

3.1 Research methods

3.1.1 Entropy-weighted TOPSIS method

In prior research on logistics evaluation, the Multi-Criteria Decision Making (MCDM) method has been extensively employed to address decision-making scenarios involving multiple criteria or attributes. Notable MCDM techniques such as the Analytic Hierarchy Process (AHP) (62), Technique for Order Preference by Similarity to Ideal Solution (TOPSIS) (63, 64), and Multi-objective Linear Programming (65) have been frequently utilized. However, the AHP often involves complex calculations and its criteria selection can introduce subjectivity. Conversely, Multi-objective Linear Programming is limited to linear problem contexts. TOPSIS, which evaluates options by considering both the negative-ideal and ideal solutions, offers a comprehensive and intuitive assessment, thus its widespread adoption. Nonetheless, the method's reliance on subjective judgment when establishing weights could potentially impact the final decision outcomes.

In response to these limitations, the Entropy Weight Method, which derives weights from the inherent variability of the data, provides a more objective alternative. The entropy-weighted TOPSIS (Technique for Order Preference by Similarity to Ideal Solution) method combines the entropy weight method and TOPSIS to objectively evaluate multidimensional decision-making problems. First, entropy weight method calculates indicator weights rooted in data dispersion, minimizing subjective bias by assigning higher weights to indicators with greater information utility. Subsequently, TOPSIS ranks alternatives by measuring their Euclidean distances to the negative and positive ideal solutions, ensuring a comprehensive assessment of relative performance. This approach offers three key advantages: (1) objectivity in weight determination through data-driven entropy values, (2) robustness in handling heterogeneous or conflicting criteria, and (3) interpretability via clear proximity scores (0-1), which quantify the gap between observed and optimal performance. In this study, it effectively captures the complexity of emergency logistics systems while avoiding arbitrary assumptions in traditional methods (e.g., AHP). This research therefore adopts the entropy-weighted TOPSIS for evaluation purposes. The initial step involves assigning weights to various measurement indicators using the Entropy Weight Method. Subsequently, TOPSIS is applied to quantitatively rank the ELRC of different provinces. The specific calculation steps are outlined as follows:

1. Standardized processing of raw data

In order to compare indicators across various provinces and years, this study utilized the Min-Max Scaling technique to normalize the data, thereby removing its dimensionality.

2. Weight calculation

Based on the entropy weighting method, this paper establishes the following weighting equations (Equations 1–3):

$$p_{\theta ij} = \frac{x'_{\theta ij}}{\sum_{\theta=1}^{r} \sum_{i=1}^{n} x'_{\theta ij}} \tag{1}$$

$$e_j = -\frac{\sum_{\theta=1}^r \sum_{i=1}^n \left(p_{\theta i j}^* \ln p_{\theta i j} \right)}{\ln rn} \tag{2}$$

$$w_{j} = \frac{1 - e_{j}}{\sum_{j=1}^{n} (1 - e_{j})}$$
(3)

Among these, there are *r* years, *n* provinces, and *m* indicators, $x_{\theta ij}$ Represents the value of the *j* indicator for the *i* province in the θ year. Here, $\theta = 1, 2, ..., r$; i = 1, 2, ..., n; $j = 1, 2, ..., m \cdot p_{\theta ij}$ represents the characteristic proportion value of the *j* indicator for the *i* province. e_j denotes the information entropy of the *j* indicator. w_i signifies the

weight of the *j* indicator, $w_j \in [0,1]$ and $\sum_{j=1}^{n} w_j = 1$.

3. Create the weighted matrix Z

$$z_{\theta ij} = w_j \times x'_{\theta ij} \left(\theta = 1, 2, \dots, r, i = 1, 2, n; j = 1, 2, \dots, m \right)$$
(4)

$$Z = \left(z_{\theta ij}\right)_{r \times m \times n} \tag{5}$$

In Equations 4, 5, $z_{\theta ij}$ represents the weighted decision score, and Z is the weighted decision matrix composed of all weighted decision scores.

4. Calculate Euclidean distance

$$d_{j}^{+} = \max(z_{\theta i j}), \ D_{\theta i}^{+} = \sqrt{\sum_{j=1}^{n} (d_{j}^{+} - z_{\theta i j})^{2}}$$
 (6)

$$d_j^{-} = \min\left(z_{\theta i j}\right), \ \mathbf{D}_{\theta i}^{-} = \sqrt{\sum_{j=1}^n \left(\mathbf{d}_j^{-} - z_{\theta i j}\right)^2} \tag{7}$$

In Equations 6, 7, d_j^+ and d_j^- respectively represent the positive and negative ideal solutions. $D_{\theta i}^+$ and $D_{\theta i}^-$ represent the Euclidean distances between the actual level of ELRC for each province and the positive and negative ideal solutions, respectively.

5. Calculate the ELRC $C_{\theta i}$ for each province:

$$C_{\theta i} = \frac{D_{\theta i}^{-}}{D_{\theta i}^{+} + D_{\theta i}^{-}} \tag{8}$$

In Equation 8, $C_{\theta i} \in [0, 1]$. The closer the value of $C_{\theta i}$ is to 1, the higher the ELRC of that region.

3.1.2 Dagum Gini coefficient

The Dagum Gini coefficient differs from the traditional Gini coefficient, coefficient of variation, and Theil index by factoring in the distribution of sub-samples. It considers issues like cross-overlap between samples (66). Consequently, the Dagum Gin coefficient effectively addresses the source of regional disparities, making it highly advantageous in analyzing spatial imbalances (67). It is commonly used to describe the issue of regional development imbalance, particularly in research related to disparities in residents' income and economic development across regions (58, 68). Additionally, in this study, it serves as a crucial tool to explore the heterogeneity of ELRC across different regions in China. The formula for calculating the Dagum Gini coefficient is shown in Equation 9:

$$G = \frac{\sum_{j=1}^{k} \sum_{h=1}^{k} \sum_{i=1}^{n_{j}} \sum_{r=1}^{n_{h}} |y_{ji} - y_{hr}|}{2\mu n^{2}}$$
(9)

Among these, *G* represents the overall Gini coefficient, *n* is the number of provinces, and *k* is the number of regions. *i* and *r* represent the number of provinces within each region. n_j (n_h) represents the number of provinces in the *j*(*h*) region. y_{ji} (y_{hr}) represents the ELRC value of the *i*(*r*) province in the *j*(*h*) region. μ represents the mean of ELRC for each province. Additionally, the Dagum Gini coefficient, using the subgroup decomposition method, allows for further breakdown of the overall Gini coefficient (*G*) into three distinct components: hypervariance density (G_{ui}), inter-regional heterogeneity (G_{uib}), and intra-regional heterogeneity (G_{wib}) (Equation 10):

$$G = G_w + G_{nb} + G_t \tag{10}$$

3.1.3 Kernel density estimation

Kernel density estimation is a widely used non-parametric method for studying the uneven distribution of samples (69). It enables the estimation of the probability density of random variables and describes the dynamic evolution trend of their distribution through continuous density curves. In this study, kernel density estimation is used to reveal the dynamic evolutionary trends of regional ELRC in China under the time dimension. In this research, the density function of regional ELRC is represented by f(x). And the probability density estimate at point x is shown in Equation 11:

$$f(x) = \frac{1}{Nn} \sum_{i=1}^{n} k \left(\frac{x_i - x}{h} \right)$$
(11)

In the above equation, *h* represents the bandwidth, *N* denotes the number of observed values, and $k(\cdot)$ represents the kernel density function. x_i represents independently and identically distributed observed values, while *x* represents the mean. This study selects the most commonly used Gaussian kernel function to investigate the dynamic evolution of China's regional ELRC distribution. The expression is shown in Equation 12:

$$K(x) = \frac{1}{\sqrt{2\pi}} \exp\left[-\frac{x^2}{2}\right]$$
(12)

3.1.4 Markov chain

The Markov chain is a stochastic process $\{X(t), t \in T\}$, where its index set *T* corresponds to different periods, and the finite types correspond to the number of types of random variables. Therefore, for all periods *t* and all types *i* and *j*, the Equation 13 should hold:

$$P\{X(t+1) = j | X(t) = i, X(t-1) = i_{t-i}, \Lambda, X(0) = i_0\}$$

= $P\{X(t+1) = j | X(t) = i\}$ (13)

Indeed, the Markov chain is commonly utilized for discretizing continuous attribute values of geographical phenomena in different periods. And, it can effectively avoid the impact of data non-temporality on prediction accuracy (70). Thus, in this research, the construction of a Markov chain is used to explore the probabilities and rules of relative transitions in China's regional ELRC. The Markov chain typically uses data categorization to divide it into *k* different types. The distribution of a certain type at time *t* is represented using a 1 × k state probability vector $E_t = [E_{1,t}, E_{2,t}, \dots, E_{k,t}]$. The entire process of state transition matrix of size $k \times k$, with probability value M_{ij} . M_{ij} denotes the probability value of a spatial unit of type *i* at time *t* transitioning to type *j* at time *t* + 1. The specific formula is shown in Equation 14:

$$M_{ij} = n_{ij} / n_i \tag{14}$$

Among these, n_{ij} represents the cumulative count of spatial units transitioning from type *i* at time *t* to type *j* at time t + 1. n_i represents the total count of spatial units of type *i* across all time steps during the research period.

3.2 Research data

Given the availability and scientific validity of the data, this study focuses on 30 provincial administrative regions, including Beijing,

TABLE 1 Comparison table of classification methods.

Classification method	Advantages	Limitations (for ELRC research)
		1. Cannot explain capability gaps within the same risk zone;
		2. Ignores the decisive role of economic foundation in disaster reduction
Disaster type classification	Reflects regional risk exposure differences	resources;
		3. Dynamic changes in disaster types (such as new risks from climate change),
		resulting in insufficient classification stability.
		1. Ignores the support of the economic hinterland for the hub functions;
Transportation hub classification	The stifter have been stated as a date	2. Difficult to quantify the emergency collaboration potential of non-hub
Transportation nub classification	identifies key logistics hodes	areas;
		3. Node importance changes dynamically with supply chain restructuring.
	1. Reveals structural differences in resource	
	allocation;	Needs to combine subgroup analysis (such as disaster types, hub levels) to
Economic region classification	2. Aligns with policy tools;	refine heterogeneity.
	3. Captures comprehensive resilience mechanisms.	

Shaanxi, Henan, Shandong, etc. (excluding the Hong Kong, Tibet Autonomous Region, Taiwan, and Macau).

The regional heterogeneity of ELRC can be categorized from multiple dimensions, such as disaster types, disaster frequency, transportation hub types, emergency evacuation functions, and economic factors. This paper conducts a regional heterogeneity analysis based on the division of China's four major economic regions for several reasons: First, the ELRC is highly dependent on long-term infrastructure investments and technological upgrades, while economic levels directly impact regional fiscal capacity and resource allocation priorities. Economically developed regions have higher fiscal budgets and can systematically build high-grade highways, among other infrastructure. Second, the division of China's four economic regions is itself embedded in national strategie, and the allocation of policy resources directly shapes regional emergency capabilities. The Eastern region attracts global logistics companies through free trade zone policies, forming an emergency network of public-private cooperation, while the western region relies on central government transfer payments, with the efficiency of fund use constrained by local governance capabilities. Third, ELRC must address complex disasters (e.g., the COVID-19 pandemic combined with extreme weather), and economic strength determines the region's system resilience under multiple shocks. A single disaster-oriented classification cannot capture such complex resilience mechanisms. Finally, the division of economic regions is highly compatible with the official statistical system, making it easier for monitoring and assessment. Economic region data is standardized, continuous, and policy-intervenable, which is more conducive to building dynamic evaluation models. In contrast, disaster types or transportation hub classifications often require cross-departmental data integration, which may present inconsistencies in scope and timeliness risks. The advantages and disadvantages of specific classification methods are shown in Table 1. In summary, this paper adopts economic regions as the basis for classification.

Simultaneously, China categorizes the entirety of its territories into four primary economic regions based on local natural conditions, economic foundations, development levels, and degrees of external openness: Eastern, Central, Western, and Northeastern. This classification aims to foster coordinated regional development. To comprehensively understand the evolving dynamics of regional ELRC, this paper refers to the economic regional classifications provided by the National Bureau of Statistics, as depicted in Figure 1. Besides, 2012

was designated as the baseline year, as it corresponds with the introduction of China's first policy aimed at developing the emergency industry, titled "Opinions on Accelerating the Development of the Emergency Industry," issued by the General Office of the State Council. Additionally, in 2018, the Ministry of Emergency Management was established in China, and corresponding emergency management departments were set up in various provinces. Due to the lag in provincial data, it is relatively difficult to obtain data after 2021. Additionally, the period from 2011 to 2021 aligns with China's 12th and 13th Five-Year Plans, during which emergency logistics policies remained relatively stable. Extending the analysis to 2023 would introduce noise from post-pandemic policy adjustments. Therefore, this study leverages observational metrics systematically documented over a ten-year continuum, commencing in 2012 and concluding in 2021. Drawing from previous research experience, a 10-year period is deemed an appropriate timeframe for analyzing the cyclical fluctuations of a subject, as it captures long-term trends and cycles while ensuring data relevance and timeliness. All original data required for this study are sourced from the website of National Bureau of Statistics² and the respective year's "China Statistical Yearbook." And some indicators are derived through calculations based on the original data, an example is the ratio of fixed asset investment in the logistics industry to the overall investment in fixed assets. For the rare cases of missing data, the mean imputation was utilized, whereby missing values were replaced by the average observed value of the respective variable.

4 Components and measurement of regional emergency logistics response capacity in China

4.1 Components and measurement indicators

Similar to regular logistics, emergency logistics consists of several key components such as flow direction, carriers, flow rate, and flow

² www.stats.gov.cn



(71). There is a consensus among scholars who study logistics capacity and, specifically, ELRC, on the measurement indicators for carrier capacity, flow rate, and velocity in logistics systems. Building on these elements, this study adopts the analytical framework for evaluating ELRC proposed by Xi (72), selecting metrics such as emergency logistics flow response capacity, emergency logistics carrier carrying capacity, emergency logistics flow velocity response capacity, regional informatization response capacity, and emergency logistics flow efficiency response capacity to identify the constitutive components of ELRC.

Furthermore, failure to incorporate contextual factors into the metric implementation of regional ELRC risks hindering capacity enhancement. As a pivotal support framework and critical safeguard mechanism for public emergencies, emergency logistics thus necessitates integrating practical considerations-particularly those arising in real-world crisis scenarios-to optimize its operational efficacy. Building on this rationale, this study posits that ensuring the availability of public medical resources during emergencies constitutes a governmental obligation across administrative tiers, while managing and coordinating emergency logistics necessitates multi-tiered governmental coordination (73-76). To operationalize this framework, Li (31) introduced social emergency security capacity as a critical dimension of ELRC. Furthermore, empirical studies identify regional informatization response capacity as a pivotal determinant of adaptation rates in logistical response systems (77). Consequently, societal emergency safeguard capacity and regional digitalization capacity are integrated as core components of ELRC. The specific composition of the indicator system is detailed in Table 2.

4.2 Analysis of measurement results

Based on the indicator system mentioned above, this study employed Stata17 software and applied the entropy-weighted TOPSIS method to calculate the ELRC of 30 provincial administrative regions in China from 2012 to 2021. And the results are presented in Table 3. Firstly, upon comparing the measurement values of each province during the period 2012-2021, a significant improvement in the ELRC is evident across all provinces in China. Secondly, from the provincial average values, it is clear that Qinghai Province exhibits the weakest ELRC, with an average value of only 0.0552. Conversely, Guangdong Province demonstrates the strongest ELRC, with a notably high average value of 0.4696. This indicates a significant disparity in ELRC among the various provinces in China. Finally, upon observing the numerical distribution of ELRC, it is evident that provinces with stronger ELRC are primarily concentrated in the Eastern region. However, provinces in the Western region display relatively weaker ELRC. This observation highlights the clear regional imbalance in China's ELRC.

After measuring and analyzing the ELRC of the four major economic regions and the entire country, the results are presented in Table 4. From a national perspective, China's regional ELRC has significantly improved during the observation period. The average across the nation increased from 0.1577 in 2012 to 0.2370 in 2021, representing a remarkable growth rate of 50.29%. The highest growth rate was observed between 2017 and 2018. It can be attributed to the establishment of the Emergency Management

TABLE 2 The index system of regional emergency logistics response capacity.

First-level indicators	Second-level indicators	Third-level indicators	Weight coefficient
	Logistics infrastructure	Transport network density	3.60%
	Logistics transportation equipment	Civil vehicle ownership	3.69%
	ownership	Ownership of road-operated vehicles	2.79%
Emergency logistics carrier carrying	Logistics carrier capacity	Express business volume	15.90%
capacity	Logistics industry practitioners	Number of employments in railway, road, aviation, handing and other transportation, warehousing, and postal services	2.33%
	Freicht volume	Road freight volume	3.03%
Emergency logistics flow response	Freight volume	Railway freight volume	8.29%
capacity	Parish (tangang	Railway freight turnover	8.16%
	Freight turnover	Road freight turnover	5.02%
		Proportion of expressways	0.77%
Emergency logistics flow velocity	Proportion of classified roads	Proportion of first-class roads	4.27%
response capacity		Proportion of second-class roads	1.28%
	Completed logistics volume per unit	Ratio of freight volume to employed personnel	2.59%
Emergency logistics flow efficiency	manpower	Ratio of freight turnover to employed personnel	1.91%
response capacity	Completed logistics volume per unit	Ratio of freight volume to logistics industry capital investment	3.18%
	capital investment	Freight turnover relative to logistics industry capital investment	6.80%
	Telecommunication business volume	Total postal and telecommunications business volume	8.20%
Regional informatization response	Network infrastructure	Optical fiber line length	3.62%
capacity		Number of mobile phone users	3.01%
	Regional communication level	Number of internet broadband access ports	3.78%
		Number of beds in health institutions	2.60%
	Pescue support foundation	Number of health technicians	2.51%
Social emergency security capacity	Rescue support foundation	Number of disease prevention and control centers	1.90%
	Regional policy support	Proportion of local fiscal transportation expenditure in total expenditure	0.78%

Department in 2018 and the subsequent release of corresponding plans. The plans emphasized the prioritization of the construction and comprehensive deployment of emergency logistics as a critical project, leading to a rapid enhancement of China's ELRC. On the regional level, China's Central, Eastern, Western, and Northeastern regions all have displayed a steady upward trend in ELRC. Among them, the Eastern region exhibited the strongest ELRC, followed by the Central and Northeastern regions, while the Western region displayed a relatively weaker capacity. Both the Eastern and Central regions surpassed the national average in ELRC. However, the Western and Northeastern regions exhibited comparatively lower capacities, falling below the national average. Moreover, from the degree of improvement in ELRC, the Eastern region had an average growth rate as high as 6.12%. The Western region experienced a growth rate of 4.18%, the Central region had a growth rate of 3.41%, and the Northeastern region had a growth rate of 1.86%. This shows that China's ELRC improvement exhibits significant regional heterogeneity.

The main reasons for the disparity in regional ELRC in China can be attributed to the following factors: firstly, China's vast territory, diverse geographical conditions, and complex climate create significant variations in natural conditions among the Western, Eastern, Central, and Northeastern regions. Consequently, these regions face different frequencies, types, and severity levels of public emergencies. Rooted in the frequency of public emergencies in China, the Eastern region experiences the most emergencies, TABLE 3 Calculation results of emergency logistics response capacities of provinces in China.

Province	2012	2013	2014	2015	2016	2017	2018	2019	2020	2021	Mean
Beijing	0.1938	0.2029	0.2033	0.2029	0.2126	0.2186	0.2252	0.2280	0.2451	0.2525	0.2185
Hebei	0.2388	0.2533	0.2656	0.2637	0.2758	0.2981	0.3211	0.3317	0.3542	0.3722	0.2975
Tianjin	0.1425	0.1299	0.1333	0.1282	0.1301	0.1340	0.1368	0.1418	0.1433	0.1435	0.1363
Shanxi	0.1793	0.1882	0.1967	0.1922	0.1941	0.2322	0.2462	0.2518	0.2579	0.2703	0.2209
Inner Mongolia	0.1908	0.1835	0.1878	0.1791	0.1834	0.2108	0.2330	0.2216	0.2180	0.2222	0.2030
Liaoning	0.2176	0.2210	0.2249	0.2258	0.2489	0.2614	0.2815	0.2705	0.2579	0.2637	0.2473
Jilin	0.1017	0.0982	0.0985	0.0951	0.0979	0.1051	0.1101	0.1128	0.1197	0.1222	0.1061
Heilongjiang	0.1348	0.1383	0.1363	0.1321	0.1359	0.1441	0.1430	0.1481	0.1465	0.1481	0.1407
Shanghai	0.2206	0.2406	0.2663	0.2540	0.2570	0.2807	0.2916	0.3220	0.3406	0.3600	0.2833
Jiangsu	0.2158	0.2487	0.2612	0.2730	0.2840	0.3066	0.3320	0.3686	0.3960	0.4274	0.3113
Zhejiang	0.1893	0.1991	0.2145	0.2348	0.2498	0.2740	0.3011	0.3354	0.3800	0.4147	0.2793
Fujian	0.1194	0.1358	0.1403	0.1470	0.1497	0.1643	0.1806	0.1892	0.1971	0.2100	0.1633
Anhui	0.2178	0.2223	0.2271	0.1980	0.2025	0.2169	0.2354	0.2270	0.2380	0.2412	0.2226
Jiangxi	0.1269	0.1359	0.1361	0.1326	0.1362	0.1561	0.1670	0.1567	0.1655	0.1714	0.1484
Shandong	0.2811	0.2795	0.2837	0.2898	0.3110	0.3250	0.3509	0.3686	0.3958	0.4133	0.3299
Henan	0.2895	0.2725	0.2858	0.2879	0.3017	0.3174	0.3380	0.3433	0.3634	0.3741	0.3174
Hubei	0.1306	0.1482	0.1566	0.1654	0.1709	0.1803	0.1939	0.1926	0.1929	0.2027	0.1734
Hunan	0.1472	0.1497	0.1576	0.1591	0.1650	0.1783	0.1901	0.1879	0.2010	0.2091	0.1745
Guangxi	0.1346	0.1397	0.1438	0.1445	0.1490	0.1609	0.1741	0.1764	0.1932	0.2022	0.1618
Guangdong	0.2887	0.3699	0.3875	0.4058	0.4310	0.4755	0.5357	0.5531	0.5954	0.6532	0.4696
Hainan	0.0683	0.0444	0.0538	0.0531	0.0516	0.0508	0.0536	0.0687	0.0855	0.0896	0.0619
Chongqing	0.0799	0.0897	0.0926	0.0960	0.0989	0.1067	0.1123	0.1141	0.1181	0.1241	0.1032
Sichuan	0.1555	0.1866	0.1964	0.2035	0.2119	0.2229	0.2486	0.2513	0.2672	0.2863	0.2230
Guizhou	0.0759	0.0849	0.0915	0.0944	0.0978	0.1045	0.1160	0.1168	0.1228	0.1305	0.1035
Yunnan	0.0993	0.1223	0.1234	0.1270	0.1305	0.1402	0.1476	0.1530	0.1612	0.1716	0.1376
Shaanxi	0.1413	0.1531	0.1656	0.1621	0.1700	0.1801	0.1901	0.1934	0.2056	0.2156	0.1777
Gansu	0.0892	0.0931	0.0931	0.0927	0.0944	0.1014	0.1120	0.1113	0.1127	0.1161	0.1016
Qinghai	0.0490	0.0482	0.0513	0.0548	0.0533	0.0520	0.0587	0.0591	0.0619	0.0638	0.0552
Ningxia	0.0981	0.0836	0.0794	0.0791	0.0798	0.0808	0.0833	0.0925	0.0991	0.0995	0.0875
Xinjiang	0.1152	0.1192	0.1199	0.1174	0.1186	0.1228	0.1408	0.1379	0.1359	0.1389	0.1267

TABLE 4 Calculation results of emergency logistics response capacities for different regions in China.

Year	Eastern region	Central region	Western region	Northeastern region	Countrywide
2012	0.1958	0.1819	0.1117	0.1514	0.1577
2013	0.2104	0.1861	0.1185	0.1525	0.1661
2014	0.2209	0.1933	0.1222	0.1532	0.1725
2015	0.2252	0.1892	0.1228	0.1510	0.1730
2016	0.2353	0.1951	0.1262	0.1609	0.1798
2017	0.2528	0.2135	0.1348	0.1702	0.1934
2018	0.2729	0.2284	0.1470	0.1782	0.2083
2019	0.2907	0.2266	0.1479	0.1771	0.2142
2020	0.3133	0.2364	0.1542	0.1747	0.2257
2021	0.3336	0.2448	0.1610	0.1780	0.2370
Mean	0.2551	0.2095	0.1346	0.1647	0.1928

followed by the Central and Western regions, while the Northeastern region faces the least incidents. Due to such differences in public emergency frequency, the accumulated emergency management experience in dealing with the incidents varies significantly across regions. Consequently, the effectiveness of emergency logistics contingency plans formulated by different regions also differs, leading to disparities in their respective ELRC. Furthermore, the lack of an established emergency information data-sharing platform and an incomplete sharing system among relevant departments contribute to n inadequate system for allocating emergency supplies across regions. Thus, the Emergency Management Department, the National Food and Strategic Reserves Administration, and other relevant departments are temporarily unable to leverage their respective strengths fully. And then, the limitation hinders the collaborative progress of different regional ELRC. Additionally, there are regional disparities in logistics infrastructure across China. The varying accessibility of emergency logistics transportation routes in different areas leads to differences in their ELRC. Lastly, differences in the scale and quality of disease control centers, the scale of medical resources, and the level of local government investment in emergency logistics collectively result in variations in social emergency security capacity among different regions. As a result, disparities in China's regional ELRC emerge.

Additionally, further examination of the indicator weights can reveal that despite leading in overall ELRC, certain provinces (e.g., Hainan) in Eastern region exhibit underperformance in "Social Emergency Security Capacity," primarily due to limited healthcare resources and fiscal expenditure allocation. The Western region showed low scores in "Transport Network Density" and "Optical Fiber Line Length" reflect infrastructure gaps, exacerbated by geographic barriers and sparse population distribution. Besides, it is a fact that moderate "Logistics Carrier Capacity" masks disparities in "Regional Informatization," where Central region (e.g., Henan) lag in digital connectivity compared to coastal counterparts.

5 Analysis of regional heterogeneity in China's emergency logistics response capacity

In this research, the Dagum Gini coefficient and its subgroup decomposition method were utilized to further analyze the overall heterogeneity of China's ELRC.

5.1 Analysis of overall heterogeneity

The overall heterogeneity can be divided into inter-regional heterogeneity, intra-regional heterogeneity, and hypervariance density. The specific calculation results are presented in Table 5.

Over the observation period, China's ELRC Gini coefficient grew from 0.2347 in 2012 to 0.2875 in 2021. It indicates a growing trend in the overall heterogeneity of China's ELRC. From the decomposition terms, both intra-regional heterogeneity and inter-regional heterogeneity demonstrated an increasing trend over the years. However, hypervariance density initially increased and then decreased. Hypervariance density and Intra-regional heterogeneity exhibited smaller values, whereas interregional heterogeneity was markedly greater. This phenomenon may be attributed to the distinct differences in economic development and logistics infrastructure across various regions in China. Despite the issuance of emergency plans and relevant policies for emergency logistics in each region, the inter-regional heterogeneity in ELRC continuously increases due to the above differences. And it contributes to further amplifying the overall heterogeneity.

5.2 Analysis of intra-regional heterogeneity

Calculating the intra-regional heterogeneity of ELRC in each economic region of China, the results are shown in Table 6. It is evident that from 2012 to 2021, intra-regional heterogeneity within various regions of China exhibited continuous fluctuations. Additionally, in certain years, intra-regional heterogeneity also displayed a cross phenomenon. Specifically, the Eastern region showed the highest level of intra-regional heterogeneity, closely followed by the Western and Northeastern regions. And the Central region exhibited the lowest level of intra-regional heterogeneity. Furthermore, both the Eastern and Western regions demonstrated overall stable intra-regional heterogeneity with a similar upward fluctuation trend, indicating a continuous increasing intra-regional heterogeneity in these regions. The intra-regional heterogeneity within the Central region remained consistently below 0.2, indicating a fluctuating downward trend. Over the period from 2012 to 2021, it declined approximately 0.0273. Conversely, the intra-regional heterogeneity within the Northeastern region exhibited three distinct phases. In the first phase, spanning from 2012 to 2015, there were continuous increases in intra-regional heterogeneity, though none surpassed 0.2. During the second phase, from 2016 to 2018, there were fluctuating increases in intra-regional heterogeneity, all exceeding 0.2. However, the third phase, covering 2019 to 2021, witnessed a trend of

TABLE 5 Overall Gini coefficient and decomposition terms of China's emergency logistics response capacities from 2012 to 2021.

					5 5 5					
Year	2012	2013	2014	2015	2016	2017	2018	2019	2020	2021
Intra-regional heterogeneity	0.0542	0.0588	0.0593	0.0596	0.0615	0.0629	0.0649	0.0644	0.0650	0.0671
Overall Gini coefficient	0.2347	0.2473	0.2510	0.2540	0.2618	0.2656	0.2696	0.2746	0.2798	0.2875
Inter-regional heterogeneity	0.1262	0.1304	0.1352	0.1391	0.1419	0.1433	0.1422	0.1556	0.1649	0.1701
Hypervariance density	0.0543	0.0581	0.0565	0.0553	0.0584	0.0594	0.0625	0.0545	0.0499	0.0503

TABLE 6 Intra-regional heterogeneity of China's ELRC from 2012 to 2021.

Year	2012	2013	2014	2015	2016	2017	2018	2019	2020	2021
Eastern region	0.1896	0.2258	0.2227	0.2281	0.2359	0.2420	0.2526	0.2468	0.2480	0.2571
Central region	0.1690	0.1408	0.1436	0.1351	0.1372	0.1307	0.1295	0.1421	0.1435	0.1417
Western region	0.1952	0.1988	0.2047	0.1996	0.2055	0.2154	0.2180	0.2105	0.2112	0.2155
Northeastern region	0.1702	0.1789	0.1833	0.1923	0.2085	0.2041	0.2137	0.1978	0.1758	0.1766

TABLE 7 Inter-regional heterogeneity of China's emergency logistics response capacity from 2012 to 2021.

Year	2012	2013	2014	2015	2016	2017	2018	2019	2020	2021
Eastern-Central	0.1934	0.2084	0.2096	0.2206	0.2302	0.2257	0.2316	0.2461	0.2550	0.2660
Eastern-Western	0.3152	0.3389	0.3431	0.3491	0.3588	0.3650	0.3649	0.3789	0.3883	0.3976
Eastern- Northeastern	0.2225	0.2609	0.2717	0.2853	0.2867	0.2954	0.3106	0.3260	0.3467	0.3634
Central-Western	0.2677	0.2489	0.2546	0.2415	0.2448	0.2528	0.2484	0.2448	0.2464	0.2450
Central- Northeastern	0.1911	0.1861	0.1967	0.2043	0.2133	0.2123	0.2260	0.2150	0.2135	0.2185
Western- Northeastern	0.2226	0.2203	0.2212	0.2192	0.2367	0.2366	0.2381	0.2235	0.2077	0.2100

diminishing intra-regional heterogeneity, with all values decreasing below 0.2. As a result, compared to other regions, the ELRC in the Northeastern region tends to be more prone to instability.

The main reason for the changes in these three phases may be that the Northeastern region, comprising the provinces of Liaoning, Heilongjiang and Jilin, serves as a significant industrial base in China. Between 2012 and 2015, China's market economy's rapid global development and continuous reforms might have instigated the relocation or shutdown of numerous traditional industrial enterprises, especially in Jilin and Heilongjiang. However, Liaoning province, benefiting from its advantageous coastal location, likely attracted greater foreign investment and the establishment of modern industries. As a derivative of the industrial sector, the logistics sector experienced an expansion in discrepancies as the industrial differences widened. Meanwhile, compared to the coastal logistics of Liaoning, Jilin and Heilongjiang rely more on their relatively underdeveloped railway network. As a result, the ELRC in Liaoning exhibited substantial variation compared to the other two provinces, leading to a persistent rise in intra-regional heterogeneity within the Northeastern region. In 2016–2018, China further propelled its supply-side structural reforms. High-energy-consuming and pollutant-intensive traditional industries faced additional constraints, causing the logistics demand in Jilin and Heilongjiang to contract further. In contrast, modern service sectors and high-tech industries received consistent encouragement and support, with logistics demands in Liaoning experiencing sustained growth. Such shifts might have intensified the disparity between Liaoning and the other two provinces, thereby elevating intra-regional heterogeneity. From 2019 to 2021, the nation actively promoted industrial upgrades and transformations to revitalize the Northeastern region. This signifies that all three Northeastern provinces received new investments and industry inflows, leading to increased logistics demands. Consequently, the ELRC across the provinces improved, indicating a trend toward reduced regional discrepancies.

5.3 Analysis of inter-regional heterogeneity

The inter-regional heterogeneity of China's ELRC was calculated. And the results are presented in Table 7. During the observation period, the highest level of heterogeneity existed between the Western and Eastern regions. However, the lowest level existed between the Northeastern and Central regions. The heterogeneity between other regions appeared between these two extremes, exhibiting a crossover phenomenon. Specifically, from 2012 to 2021, the heterogeneity between the Western and Eastern regions exceeded 0.3, indicating significant disparities between these regions. In opposition, the heterogeneity between the Central and Northeastern regions fluctuated around 0.2, reflecting relatively more minor differences. Likewise, the heterogeneity between the Western and Central regions and between the Northeastern and Western regions showed comparable levels, hovering around 0.25 and 0.22, respectively. Notably, the heterogeneity between the Eastern and other regions exhibited an upward trend over time. This may be directly attributed to the Eastern region's developed economic conditions and advanced logistics infrastructure. However, it also highlights the evident unevenness in the ELRC across China's Northeastern, Western, Central, and Eastern regions.

5.4 Analysis of the sources of regional heterogeneity

The impact levels of the Dagum Gini coefficient's decomposition terms were calculated. Figure 2 displays the results, with the average contribution rate of inter-regional heterogeneity being the highest, at 55.08%. The average contribution levels of intra-regional heterogeneity is 23.52%. In comparison, the mean contribution rate of hypervariance density is the lowest, at only 21.40%. This shows that the overall heterogeneity in ELRC mainly



stems from inter-regional heterogeneity. However, intra-regional heterogeneity and hypervariance density have similar and relatively lower contribution rates, exerting little influence on the overall heterogeneity. The specific reasons may lie in the differences in regional resource endowments and resource allocation. Developed Eastern region enjoys distinct locational advantages and possesses higher levels of economic development and infrastructure. By contrast, other regions lag in various production factors. Consequently, capital investment is more inclined to flow into the Eastern region, significantly boosting its ELRC. Regarding the trends, the contribution rate of hypervariance density fluctuated up and down from 2012 to 2018 but declined from 2019 to 2021. In contrast, both intra-regional and inter-regional heterogeneity contribution rates showed an increasing trend annually.

6 Spatiotemporal evolution characteristics of China's regional emergency logistics response capacity

6.1 Dynamic evolution characteristics in the time dimension

This study utilized kernel density estimation to depict the timebased changes in China's regional ELRC.

6.1.1 Dynamic evolution characteristics at the national level

The dynamic evolution characteristics of ELRC under the time dimension at the national level are presented in Figure 3. From observing the distribution positions, the overall centre of the ELRC distribution curve gradually shifted to the right during the observation period, indicating effective improvement in China's ELRC. Regarding the distribution shape, there was only one peak during the observation period. And the values were mainly concentrated between 0 and 0.2. This seems that China's ELRC remained relatively stable during the observation period without significant polarization. Additionally, the main peak's height gradually decreased, and its width became wider. This implies that the measured values of ELRC showed a trend of gradual dispersion during the observation period. In other words, the disparities between different regions' ELRC have been gradually widening. The stretching of the curve shows that, as the years progressed, the right-tail phenomenon of the kernel density curve decreased. Especially in 2020 and 2021, the kernel density curve showed an upward trend and data aggregation at certain high-level positions. This could be attributed to the widespread outbreak of COVID-19 and the frequent occurrence of severe autumn floods in the middle and lower reaches of the Yellow River, hail disasters in the Eastern region, and snow disasters in the northern part of China. The capability of governments at all levels to respond to and dispatch resources during emergencies has been continually enhanced, subsequently elevating the regional ELRC.

6.1.2 Dynamic evolution characteristics at the economic region level

Through the kernel density curve analysis of the ELRC in four major economic regions of China, the results are plotted in Figure 4. The kernel density curves of every region displayed a shift to the right throughout the observation period. This indicates significant improvements in the ELRC of the Eastern, Northeastern, Western, and Central regions. Specifically, the highest ELRC show up in Eastern region, with the peak values of the curve clustering around 0.3. The Central and Northeastern regions followed, with peak values gathering around 0.2 and 0.15, respectively. Lastly, the Western region had the lowest ELRC, with the peak of the kernel density curve ranging between 0.1 and 0.15. Furthermore, both the Western and Eastern regions' distribution curves showed a gradual decrease in the heights of the main peaks, and their widths became wider. This indicates that the internal differences in ELRC within the Eastern and Western regions have been expanding, which corresponds to the conclusions taken from the Dagum Gini coefficient. Moreover, from the perspective of curve ductility, the right-tail phenomenon of the kernel density curves for the Western and Eastern regions showed a decreasing trend with reduced convergence. This indicates a significant growth trend in these two regions' ELRC, which is consistent with the analysis presented earlier.



6.2 Evolution characteristics in the spatial dimension

In order to better reveal the evolving trends and particular transfer characteristics of China's ELRC, this paper used the Markov transition probability matrix to analysis. Considering the distinctive features of the research area, the observed ELRC during the sample period was classified into four types using the standard quartile method: poor-level type (I), moderate-level type (II), good-level type (III), and excellent-level type (IV). MATLAB R2021a software was utilized to calculate both the conventional and spatial Markov chain transition probability matrices. The results are presented in Tables 8, 9.

6.2.1 Traditional Markov chain analysis

According to the traditional Markov chain transition probability matrix exhibited in Table 8, from 2012 to 2021, the retention probabilities for the four types of China's regional ELRC (i.e., poor-level region, moderate-level region, good-level region, and excellent-level region) were 92.9, 91.2, 88.2, and 100%, respectively. These results indicate that China's regional ELRC maintained stability in its original state during the observation period. Particularly, the excellent-level region had a retention probability of 100%, demonstrating strong stability in its ELRC, consistently maintaining an excellent-level performance. Moreover, the ranking of retention probabilities from highest to lowest is as follows: excellent-level region, poor-level region, moderate-level region, and good-level region. This suggests that China's overall ELRC is transitioning from a poor-level to a better level, with a trend of continuous improvement. However, the regions with poorer capacity are more likely to remain in their original states, and regions with better capacity tend to maintain their existing ELRC levels. This exhibits specific characteristics of the Matthew effect, likely due to the imperfect logistics infrastructure and emergency event management systems in regions with lower capacity. From the perspective of transition probabilities, there is a 7.1% probability of the ELRC in the poor-level region transitioning to the moderate-level. The moderate-level region has an 8.8% probability of transitioning to the good-level, while the good-level region has an 11.8% probability of transitioning to the excellent-level. It can be observed that the improvement of ELRC occurs only between adjacent levels. And there is no cross-level transition. This appears that the enhancement of China's ELRC is continuous, and achieving a significant leap in a short period is challenging.

6.2.2 Spatial Markov chain analysis

Based on the traditional Markov chain, this study introduced a spatial weight matrix to incorporate the "spatial lag" factor, thus establishing a spatial Markov chain. By analyzing the spatial Markov chain transition probability matrix presented in Table 9, several inferences can be drawn.

Firstly, the transfer probability matrices differed across various spatial lag types. This indicates that, in the context of heterogeneous ELRC among neighbouring provinces, the probability of ELRC undergoing transfer varies accordingly within this province. It further suggests that spatial factors exert a certain influence on ELRC. Furthermore, the transfer probability matrices exhibit an inclination trend in the upper-right triangular region, revealing a rising pattern of China's ELRC in the spatial dimension.

Secondly, when the ELRC of the local region is at a moderate-level and neighbouring regions is at a poor-level, the probability of transitioning from the moderate-level to the good-level is 20%. It is significantly higher than the traditional Markov chain's 8.8%. This suggests that, in the spatial dimension, when adjacent regions have poorer levels, the local region is more inclined to undergo a betterlevel leap. This might be attributed to the local region's ability to attract specific emergency logistics resources from neighbouring lower-level areas, thereby elevating its capacity.

Thirdly, when the local regional ELRC reaches an excellent-level, it remains at the excellent-level, irrespective of the level of the adjacent areas. This demonstrates that provinces with excellent ELRC exhibit strong stability. And they are not easily affected by the spatial correlation effects from neighbouring provinces with relatively lower ELRC.

Fourthly, when the ELRC in an adjacent region remains at a poorlevel, regions at poor-level, moderate-level, and good-level will exert a more evident absorptive effect on it. For instance, when the ELRC in neighbouring regions is at a poor-level, and the local region also operates at a poor-level, the rate of the local region transitioning from a poor-level to a moderate-level is 14.3%. It significantly exceeds the





conventional Markov chain's probability of 7.1%. This indicates a substantial increase in the likelihood of transitioning from a poorlevel region to a moderate-level region. This may be because regions with relatively better or comparable ELRC tend to attract resources from regions operating at a poor-level. Thereby, it exerts positive spatial effects on their own ELRC development. Similarly, when the ELRC in adjacent regions reaches a good-level, the rate of the local region transitioning from a poor-level to a moderate-level is 13.3%. It also far surpasses the conventional Markov chain's probability of 7.1%. This can be attributed to the implementation of supportive emergency logistics policies by local governments and the directed allocation of resources toward regions with relatively poorer ELRC. These measures act as positive driving forces for enhancing the ELRC of areas operating at a poor-level.

TABLE 8 Traditional Markov chain transition probability matrix of China's emergency logistics response capacities from 2012 to 2021.

Туре	I	II	111	IV	Observation value
Ι	0.929	0.071	0.000	0.000	70
II	0.000	0.912	0.088	0.000	68
III	0.000	0.000	0.882	0.118	68
IV	0.000	0.000	0.000	1.000	64

TABLE 9 Spatial Markov chain transition probability matrix of China's emergency logistics response capacities from 2012 to 2021.

Domain type	Туре		II	III	IV	Observation value
	Ι	0.857	0.143	0.000	0.000	7
T	II	0.000	0.800	0.200	0.000	5
1	III	0.000	0.000	1.000	0.000	5
	IV	0.000	0.000	0.000	1.000	2
	Ι	0.949	0.051	0.000	0.000	39
	II	0.000	0.962	0.038	0.000	26
11	III	0.000	0.000	0.875	0.125	8
	IV	0.000	0.000	0.000	1.000	9
	Ι	0.867	0.133	0.000	0.000	15
	II	0.000	0.923	0.077	0.000	26
111	III	0.000	0.000	0.886	0.114	44
	IV	0.000	0.000	0.000	1.000	24
	Ι	1.000	0.000	0.000	0.000	9
	II	0.000	0.818	0.182	0.000	11
IV	III	0.000	0.000	0.818	0.182	11
	IV	0.000	0.000	0.000	1.000	29

Fifthly, when the ELRC in adjacent regions reaches an excellentlevel, the percentage of the local region transitioning from a poorlevel to a moderate-level is 0%. It is significantly lower than the traditional Markov chain's probability of 7.1%. This appears that the excellent-level ELRC in neighbouring regions negatively impacts regions operating at a poor-level, making it challenging for them to achieve rapid improvements in their ELRC. Meanwhile, the percentage of the local region transitioning from a moderate-level to a good-level is 18.2%, higher than the traditional Markov chain's probability of 8.8%. Similarly, the percentage of the local region advancing from a good-level to an excellent-level is 18.2%, also surpassing the traditional Markov chain's probability of 11.8%. This reveals that, under the circumstance of adjacent regions with an excellent-level ELRC, regions operating at a moderate-level and a good-level are positively influenced, making it easier for them to undergo transformative upgrades. This could be attributed to the weaker foundations in regions with poor-level ELRC, such as information technology and emergency logistics facilities. Consequently, the radiative driving effect from neighbouring regions with excellent-level capacity is restricted, leading to an increase in spatial disparities in ELRC. In contrast, areas with a certain level of foundation at moderate and good levels are more susceptible to the radiative driving effect from neighbouring regions with excellentlevel capacity. This makes it easier for their ELRC to achieve transformative upgrades.

Lastly, even when neighbouring excellent-level regions, the rate of transitioning from a poor-level region to a good-level region remains at 0. And the rate of a transition from a moderate-level region to an excellent-level region also remains at 0. This seems that transitions only occur between adjacent levels of ELRC. A likely explanation is that building emergency logistics infrastructure and enhancing management systems are gradual processes, making significant improvements difficult in a short time. Therefore, such a gradual progression hinders the rapid transformative upgrades of regional ELRC.

7 Discussion

This study systematically evaluates the heterogeneity and spatiotemporal evolution of regional ELRC in China through a multidimensional index system and advanced analytical methods. The findings yield critical theoretical and practical implications, while also addressing gaps in existing literature on emergency logistics governance. Below, we synthesize the core insights, contextualize them within broader academic debates, and outline pathways for future research. The central findings can be distilled into three core insights: persistent expansion of regional heterogeneity, gradual progression coupled with stability in spatiotemporal evolution, and dual effects inherent to spatial interdependence.

7.1 Key contributions to theory and practice

First, our analysis reveals a persistent and widening spatial disparity in ELRC across China, characterized by a pronounced "East–West gradient." The Dagum Gini coefficient decomposition demonstrates that inter-regional heterogeneity (contributing 55.08% to overall inequality) dominates this pattern, particularly between the economically advanced Eastern region and underdeveloped Western provinces. This aligns with the "core-periphery" theory in regional economics (78), where resource concentration in core areas exacerbates spatial imbalances.

Second, unlike prior studies focusing on static evaluations (27, 29), our spatiotemporal approach uncovers the self-reinforcing nature of regional disparities: high-capacity regions (e.g., Guangdong) exhibit strong stability (100% retention probability), while low-capacity regions (e.g., Qinghai) face structural barriers to improvement, reflecting a "Matthew effect" in emergency resource allocation.

Third, the study identifies dual spatial spillover effects shaping ELRC evolution. Spatial Markov chain analysis highlights that proximity to high-capacity regions can both facilitate and hinder local capacity development. For instance, moderate-capacity regions adjacent to high-capacity neighbors show an 18.2% probability of upgrading (vs. 8.8% in isolation), suggesting technology diffusion and policy emulation. Conversely, low-capacity regions bordering high-capacity areas experience suppressed upgrading probabilities (0% for low to moderate transitions), likely due to resource siphon effects. This duality echoes debates on "trickle-down" versus "polarization" effects in regional development (79), offering empirical evidence specific to emergency logistics systems.

7.2 Heterogeneity and influencing factors

The heterogeneity in China's regional ELRC is not only a reflection of spatial disparities but also a product of multifaceted influencing factors. Our analysis reveals that resource endowments, governance capacity, and spatial spillovers collectively shape the uneven distribution of ELRC across regions.

First, this study conjects that the resource endowment gap between the Eastern and Western regions is a primary driver of heterogeneity. The Eastern region benefits from advanced logistics infrastructure (e.g., high-grade highways, dense railway networks) and robust digital connectivity (e.g., fiber-optic coverage), which are critical for rapid disaster response. In contrast, the Western region's rugged terrain and underdeveloped transportation networks hinder the efficient allocation of emergency resources. This aligns with the findings of Chen et al. (13), who highlighted the role of geographical constraints in exacerbating regional imbalances in logistics development.

Second, governance capacity plays a pivotal role in shaping ELRC. Provinces with higher fiscal expenditure on transportation and healthcare (e.g., Guangdong, Jiangsu) demonstrate stronger emergency preparedness and response capabilities. This is consistent with the literature on disaster management (75), which emphasizes the importance of institutional frameworks in enhancing resilience. However, our study extends this perspective by revealing that policy implementation gaps—such as uneven enforcement of emergency plans and limited interdepartmental coordination—further amplify regional disparities. Finally, socioeconomic factors such as population density, industrial structure, and climate risks also contribute to heterogeneity. For example, the Eastern region's high population density and exposure to frequent disasters (e.g., typhoons, floods) have driven investments in resilient infrastructure and emergency management systems. In contrast, the Northeastern region's reliance on traditional industries and aging population has limited its capacity to adapt to emerging challenges. These findings echo the work of Zhou et al. (2), who identified socioeconomic vulnerabilities as key determinants of disaster resilience.

7.3 Limitations and future research directions

While providing novel insights, this study has limitations that warrant further exploration:

- Due to time difficulties and constraints in data acquisition, the 2012–2021 dataset predates recent mega-disasters (e.g., 2023 Henan floods), limiting insights into post-pandemic resilience dynamics. Future work should incorporate real-time disaster response data to validate the index system's predictive validity.
- 2. It has not delved into the exploration of influencing factors on ELRC, especially the role of factors such as resource endowments, geographical landscapes, logistics infrastructure, and climate. Therefore, future research can consider incorporating the study and analysis of the influencing mechanisms on regional ELRC to provide theoretical support and decision-making references for enhancing ELRC.
- 3. China's centralized governance model may limit the transferability of findings to decentralized systems. Comparative studies across political regimes (e.g., U.S. federal vs. EU supranational models) could identify context-specific optimization strategies.

Future research should prioritize two avenues:

- 1. Integrating Geographic Information Systems (GIS) and machine learning algorithms to analyze the impacts of dynamic factors—such as climate change and population mobility—on ELRC. For example, simulate resource flows under stochastic disaster scenarios to stress-test regional coordination protocols. Or, embed climate risk projections into ELRC assessments to address escalating environmental uncertainties.
- Explore synergies between emergency logistics networks and sustainable development goals (SDGs), particularly SDG 9 (infrastructure) and SDG 11 (resilient cities).

8 Conclusions and implications

8.1 Research conclusions

The research constructed a comprehensive index system for regional ELRC and applied the entropy-weighted TOPSIS method to measure the ELRC of 30 provincial-level administrative regions in China from 2012 to 2021. Subsequently, the Dagum Gini coefficient, its subgroup decomposition, Markov chain and kernel density estimation, analysis were used to conduct heterogeneity and spatiotemporal evolution analyzes. The conclusions achieved are as follows.

From 2012 to 2021, the ELRC of various provinces in China have shown significant improvement. Particularly, provinces with better ELRC are mainly concentrated in the Eastern regions. In contrast, provinces in the western regions exhibit relatively poorer levels of ELRC.

The overall Gini coefficient for China's ELRC presents a continuous expansion trend, and the heterogeneity of ELRC is mainly derived from inter-regional heterogeneity. Moreover, the ELRC in the Central, Eastern, Western, and Northeastern regions have been steadily improved. No polarization phenomenon was observed, and the performance appeared relatively stable. However, the regional disparities have gradually expanded over time. The analysis using the Markov chain reveals that China's ELRC is undergoing a gradual transition from a poorer level to a better level, showing a continuous upward trend. However, at the same time, it exhibits specific characteristics of the Matthew effect.

8.2 Policy implications

Drawing from the research conclusions above, this paper offers the following policy recommendations:

The first is targeted infrastructure investment. For the Western region, prioritize high-grade highway construction (e.g., Sichuan-Qinghai corridors) and subsidize rural optical fiber deployment to bridge digital divides. Modernize railway freight networks in Northeastern region and incentivize public-private partnerships to upgrade aging transport fleets. Fully harness market resources, refine the mainline transportation and regional distribution systems of emergency logistics and augment the ELRC to various unexpected incidents.

The second is inter-regional collaborative mechanism development. Establish East–West Resource Redistribution Platforms, such as leveraging Guangdong's advanced ELRC to mentor western provinces (e.g., Qinghai) by sharing emergency reserves and conducting joint training programs. Besides, construct a unified data-sharing platform for the visualization, dissemination, and coordinated command of emergency logistics information. Especially in the Central region, develop cross-provincial data hubs (e.g., Wuhan-Xi'an) to enhance realtime information sharing during multi-hazard events.

The third is dynamic performance monitoring framework. Implement a tiered evaluation system to track ELRC advancements. Provinces achieving tier transitions (e.g., low-to-medium) should receive fiscal incentives, while underperforming regions must be mandated to develop corrective action plans. Concurrently, instate inter-regional technological exchange and collaboration platforms, promoting the sharing and dissemination of best practices. Encourage enterprises and institutions in the Eastern region to foster

References

1. Zhou L, Wu XH, Xu ZS, Fujita H. Emergency decision making for natural disasters: an overview. Int J Disaster Risk Reduct. (2018) 27:567–76. doi: 10.1016/j.ijdrr.2017.09.037

2. Zhou Y, Liu YS, Wu WX, Li N. Integrated risk assessment of multi-hazards in China. *Nat Hazards*. (2015) 78:257-80. doi: 10.1007/s11069-015-1713-y

technological collaboration and exchanges with their Western counterparts. Furthermore, the government should incentivize logistics enterprises to invest and expand in the Western region through tax incentives or fiscal subsidies.

Data availability statement

Publicly available datasets were analyzed in this study. This data can be found here: "China Statistical Yearbook" and the website of National Bureau of Statistics (www.stats.gov.cn).

Author contributions

HC: Conceptualization, Formal analysis, Funding acquisition, Investigation, Project administration, Resources, Supervision, Validation, Writing – original draft, Writing – review & editing. XL: Data curation, Formal analysis, Investigation, Methodology, Software, Visualization, Writing – original draft, Writing – review & editing. YG: Formal analysis, Methodology, Software, Visualization, Writing – original draft, Writing – review & editing. XQ: Software, Visualization, Writing – original draft, Writing – review & editing.

Funding

The author(s) declare that financial support was received for the research, authorship, and/or publication of this article. This research is funded by the National Social Science Fund General Project "Evaluation, Impact Mechanism and Policy Improvement of China's Regional Emergency Logistics Rapid Response Capability under Public Emergencies", grant number 22BJY159.

Conflict of interest

The authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

Publisher's note

All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

^{3.} Lu XL, Xue L. Managing the unexpected: sense-making in the Chinese emergency management system. *Public Adm.* (2016) 94:414–29. doi: 10.1111/padm.12261

^{4.} Zhou Z, Jin Y, Zeng W. Talking About the Status Quo of China's Emergency Logistics in Major Disasters—A Case Study of Jiuzhaigou Earthquake. In: V Sugumaran, ZPS Xu and H Zhou. (eds) Application of Intelligent Systems in Multi-modal

Information Analytics. MMIA 2019. Advances in Intelligent Systems and Computing. Cham: Springer Nature, (2019) 929. doi: 10.1007/978-3-030-15740-1_66

5. da Costa SRA, Campos VBG, de Mello Bandeira RA. Supply chains in humanitarian operations: cases and analysis. *Procedia Soc Behav Sci.* (2012) 54:598-607. doi: 10.1016/j.sbspro.2012.09.777

6. Chen ZJ, Xu L, Cheng Y. Discussion on the reliability of emergency logistics supply chain. *Railw Transp Econ.* (2009) 31:69–71. doi: 10.3969/j.issn.1003-1421. 2009.09.021

7. Xu ZY, Ren SK, Guo XS, Yuan ZP. Reliability evaluation model of emergency supply chain under uncertainty. *Oper Res Manag Sci.* (2015) 24:35–44. doi: 10.12005/ orms.2015.0081

8. Lin YM, Zhuo JX, Yang K. Research on emergency logistics capability evaluation based on multi-level gray evaluation method. *Logist Sci-Tech.* (2010) 33:57–60. doi: 10.3969/j.issn.1002-3100.2010.10.017

9. Deng AM, Zhang F, Xiong J, Wu PF. Evaluation of emergency logistics capability based on fuzzy Gray Comprehensive evaluation method. *Stat Decis*. (2010) 306:174–6. doi: 10.13546/j.cnki.tjyjc.2010.06.010

10. Banomyong R, Sopadang A. Using Monte Carlo simulation to refine emergency logistics response models: a case study. *Int J Phys Distrib Logist Manag.* (2010) 40:709–21. doi: 10.1108/09600031011079346

11. Hadiguna RA, Kamil I, Delati A, Reed R. Implementing a web-based decision support system for disaster logistics: a case study of an evacuation location assessment for Indonesia. *Int J Disaster Risk Reduct.* (2014) 9:38–47. doi: 10.1016/j.ijdrr.2014.02.004

12. Salam MA, Khan SA. Lessons from the humanitarian disaster logistics management: a case study of the earthquake in Haiti. *BIJ.* (2020) 27:1455–73. doi: 10.1108/BIJ-04-2019-0165

13. Chen H, Wei XJ, Yin XY. The unbalanced development of China's logistics industry and its stage characteristics—based on the perspective of labor input. *J Quant Technol Econ.* (2016) 33:3–22. doi: 10.13653/j.cnki.jqte.2016.11.001

14. Ou ZW, Wang HY, Jiang DL, Lu BL, Gan WX, Liang J. Emergency logistics. J Chongqing Univ. (2004) 3:164–7. Available at: http://qks.cqu.edu.cn/cqdxzrcn/article/abstract/200403124

15. Wang XY, Chen HQ, Wang KQ. Studies on emergency logistics operation model for unexpected events at Yangtze chemical industrial park. *Procedia Eng.* (2012) 43:353–8. doi: 10.1016/j.proeng.2012.08.061

16. Xiong G.J. Study on model building of emergency logistics management information system. 2010 International Conference on Logistics Engineering and Management (ICLEM 2010). Chengdu, China: IEEE (2010) 199–201.

17. Chang FS, Wu JS, Lee CN, Shen HC. Greedy-search-based multi-objective genetic algorithm for emergency logistics scheduling. *Expert Syst Appl.* (2014) 41:2947–56. doi: 10.1016/j.eswa.2013.10.026

18. Yáñez-Sandivari L, Cortés CE, Rey PA. Humanitarian logistics and emergencies management: new perspectives to a sociotechnical problem and its optimization approach management. *Int J Disaster Risk Reduct.* (2021) 52:101952. doi: 10.1016/j.ijdrr.2020.101952

19. Sun X.L., Lu J. In a method of emergency logistics route choice based on fuzzy theory. 2010 International Conference on Logistics Engineering and Management (ICLEM 2010). Chengdu, China: ASCE. (2010) 2813–2819.

20. Yi W, Kumar A. Ant colony optimization for disaster relief operations. *Transp Res E Logist Transp Rev.* (2007) 43:660–72. doi: 10.1016/j.tre.2006.05.004

21. Jiang YP, Yuan YF. Emergency logistics in a large-scale disaster context: achievements and challenges. *Int J Environ Res Public Health*. (2019) 16:779. doi: 10.3390/ijerph16050779

22. Huang XX, Song LY. An emergency logistics distribution routing model for unexpected events. Ann Oper Res. (2018) 269:223–39. doi: 10.1007/s10479-016-2300-7

23. Kundu T, Sheu JB, Kuo HT. Emergency logistics management—review and propositions for future research. *Transp Res E Logist Transp Rev.* (2022) 164:102789. doi: 10.1016/j.tre.2022.102789

24. Kołodziejczyk P, Szołtysek J. Epistemology of social logistics. Organ Rev. (2009) 4:21-4. doi: 10.33141/po.2009.04.06

25. Lei L. A preliminary study on emergency logistics. *Stat Decis*. (2004) 6:122–3. doi: 10.3969/j.issn.1002-6487.2004.06.069

26. Caunhye AM, Nie XF, Pokharel S. Optimization models in emergency logistics: a literature review. *Socio Econ Plan Sci.* (2012) 46:4–13. doi: 10.1016/j.seps.2011.04.004

27. Sheu JB. An emergency logistics distribution approach for quick response to urgent relief demand in disasters. *Transp Res E Logist Transp Rev.* (2007) 43:687–709. doi: 10.1016/j.tre.2006.04.004

28. Carter WN. Major requirements for coping with disaster In: Disaster management A disaster manager's handbook. Mandaluyong City, Philippines: Asian Development Bank (2008). 35–41.

29. Zhang YC, Ding QY, Liu JB. Performance evaluation of emergency logistics capability for public health emergencies: perspective of COVID-19. *Int J Log Res Appl.* (2022) 25:1509–22. doi: 10.1080/13675567.2021.1914566

30. Zhang BB, Lu WJ. National highway trunk line penetration and corporate inventory adjustment: Evidence from manufacturing firms in the western region. *JCUFE*. (2018) 10:114–28. doi: 10.19681/j.cnki.jcufe.2018.10.011

31. Li XD, Wang YQ, Wang F. Research on the improvement of emergency logistics based on blockchain application under public health emergencies. *Contemp Econ Manag.* (2020) 42:57–63. doi: 10.13253/j.cnki.ddjjgl.2020.04.009

32. Wang YH, Wu JX. Research on the optimization of emergency logistics path under sudden public security incidents based on cloud platform. *J Technol Econ*. (2021) 40:162–73. doi: 10.3969/j.issn.1002-980X.2021.12.015

33. Rodríguez-Espíndola O, Chowdhury S, Beltagui A, Albores P. The potential of emergent disruptive technologies for humanitarian supply chains: the integration of blockchain, artificial intelligence and 3D printing. *Int J Prod Res.* (2020) 58:4610–30. doi: 10.1080/00207543.2020.1761565

34. Liu YW, Li L, Tu Y, Mei YL. Fuzzy TOPSIS-EW method with multi-granularity linguistic assessment information for emergency logistics performance evaluation. *Symmetry*. (2020) 12:1331. doi: 10.3390/sym12081331

35. Wang Y, Peng SH, Xu M. Emergency logistics network design based on space-time resource configuration. *Knowl-Based Syst.* (2021) 223:107041. doi: 10.1016/j.knosys. 2021.107041

36. Chen TG, Wu SW, Yang JJ. Risk propagation model and its simulation of emergency logistics network based on material reliability. *Int J Environ Res Public Health.* (2019) 16:4677. doi: 10.3390/ijerph16234677

37. Balcik B, Beamon BM. Facility location in humanitarian relief. Int J Logist. (2008) 11:101–21. doi: 10.1080/13675560701561789

38. Feng JR, Gai WM, Li JY. Location selection of emergency supplies repositories for emergency logistics management: a variable weighted algorithm. *J Loss Prev Process Ind.* (2020) 63:104032. doi: 10.1016/j.jlp.2019.104032

39. Wang W, Wu SN, Wang SA, Zhen L, Qu XB. Emergency facility location problems in logistics: status and perspectives. *Transp Res E Logist Transp Rev.* (2021) 154:102465. doi: 10.1016/j.tre.2021.102465

40. Lai ZZ, Wang Z, Ge DM, Chen YL. A robust optimization model for site selection of multi-objective emergency logistics center. *Oper Res Manag Sci.* (2020) 29:74–83. doi: 10.12005/orms.2020.0121

41. Yuan Y, Wang DW. Path selection model and algorithm for emergency logistics management. *Comput Ind Eng.* (2009) 56:1081–94. doi: 10.1016/j.cie.2008.09.033

42. Tuzkaya UR, Heragu SS, Evans GW, Johnson M. Designing a large-scale emergency logistics network–a case study for Kentucky. *Eur J Ind Eng.* (2014) 8:513–32. doi: 10.1504/EJIE.2014.064758

43. Liu HB, Song GP, Liu TY, Guo B. Multitask emergency logistics planning under multimodal transportation. *Mathematics*. (2022) 10:3624. doi: 10.3390/math10193624

44. Sun HL, Xiang MK. Research on uncertain demand emergency location-path robust optimization under the risk of facility failure. *Chin J Manag Sci.* (2020) 28:199–207. doi: 10.3969/j.issn.1005-2542.2019.06.013

45. Sheng HY, Liu CS, Lu RY. Research on the positioning-routing problem under the situation of shortage of emergency supplies in the early stage after the earthquake. *Oper Res Manag Sci.* (2019) 28:41–7. doi: 10.12005/orms.2019.0126

46. Peng ZX, Wang C, Xu WQ, Zhang JS. Research on location-routing problem of maritime emergency materials distribution based on bi-level programming. *Mathematics*. (2022) 10:1243. doi: 10.3390/math10081243

47. Özdamar L, Ekinci E, Küçükyazici B. Emergency logistics planning in natural disasters. Ann Oper Res. (2004) 129:217–45. doi: 10.1023/B:ANOR.0000030690.27939.39

48. Garrido RA, Lamas P, Pino FJ. A stochastic programming approach for floods emergency logistics. *Transp Res E Logist Transp Rev.* (2015) 75:18–31. doi: 10.1016/j.tre.2014.12.002

49. Qu CC, Wang J, Yu JH. Research on decision-making model and algorithm of post-disaster emergency logistics based on grid management. *Oper Res Manag Sci.* (2022) 31:75–9. doi: 10.12005/orms.2022.0011

50. Wang HM, Xu RF, Zijie X, Zhou XG, Wang QH, Duan Q, et al. Research on the optimized dispatch and transportation scheme for emergency logistics. *Procedia Comput Sci.* (2018) 129:208–14. doi: 10.1016/j.procs.2018.03.043

51. Yuan Y, Wang F. An emergency supplies scheduling for chemical industry park: based on super network theory. *Environ Sci Pollut Res.* (2022) 29:39345–58. doi: 10.1007/s11356-021-18182-y

52. Gong BG, Chen X, Hu CZ. Fuzzy entropy clustering approach to evaluate the reliability of emergency logistics system. *Energy Procedia*. (2012) 16:278–83. doi: 10.1016/j.egypro.2012.01.046

53. Zhang YZ, Zeng HF, Nie Y. Research on reliability evaluation of urban emergency logistics system. *Logist Sci-Tech*. (2014) 37:28–31. doi: 10.13714/j.cnki.1002-3100.2014.04.008

54. Liu M, Cao J, Zhang D. Dynamic adjustment and optimization of data-driven epidemic emergency logistics network. *Syst Eng Theory Pract.* (2020) 40:437–48. doi: 10.12011/1000-6788-2018-1690-12

55. Ji JH, Zhang RF. Performance evaluation of grain emergency logistics system based on extensible matter-element model. *Logist Technol.* (2014) 33:130–3. doi: 10.3969/j. issn.1005-152X.2014.01.041

56. Zhang NP, Yao Y, Chen L, Wang Z. Research on performance evaluation method of emergency logistics based on evidence reasoning. *Logist Sci-Tech*. (2016) 39:32–7. doi: 10.13714/j.cnki.1002-3100.2016.07.009

57. Li DP, Wang YB, Cui GQ, Lu LL. Research on performance evaluation of emergency logistics system based on MDHGF. *J Shanxi Univ Financ Econ.* (2012) 34:110–1. doi: 10.13781/j.cnki.1007-9556.2012.s3.139

58. Zhang LY, Ma X, Ock YS, Qing LL. Research on regional differences and influencing factors of Chinese industrial green technology innovation efficiency based on dagum gini coefficient decomposition. *Land.* (2022) 11:122. doi: 10.3390/land11010122

59. Yang YX, Guo ZX, He ZF. Multi-attribute decision making method based on probabilistic linguistic term sets and its application in the evaluation of emergency logistics capacity. *J Intell Fuzzy Syst.* (2022) 42:2157–68. doi: 10.3233/JIFS-211495

60. Sun J, Tan QM. Evaluation of emergency logistics capability based on GF-AHP combined model—taking earthquake disasters as an example. *J Beijing Univ Technol.* (2014) 40:1354–60. doi: 10.11936/bjutxb2014091354

61. Chen JH, Ke XY. Evaluation of emergency logistics capability based on DEA-fuzzy comprehensive evaluation method. *Logist Technol.* (2015) 34:114–6. doi: 10.3969/j. issn.1005-152X.2015.01.037

62. Multaharju S, Hallikas J. Logistics service capabilities of logistics service provider. Int J Logist Syst Manag. (2015) 20:103–21. doi: 10.1504/IJLSM.2015.065975

63. Kannan G, Pokharel S, Kumar PS. A hybrid approach using ISM and fuzzy TOPSIS for the selection of reverse logistics provider. *Resour Conserv Recycl.* (2009) 54:28–36. doi: 10.1016/j.resconrec.2009.06.004

64. Bottani E, Rizzi A. A fuzzy TOPSIS methodology to support outsourcing of logistics services. *Supply Chain Manag.* (2006) 11:294–308. doi: 10.1108/13598540610671743

65. Zarbakhshnia N, Soleimani H, Goh M, Razavi SS. A novel multi-objective model for green forward and reverse logistics network design. *J Clean Prod.* (2019) 208:1304–16. doi: 10.1016/j.jclepro.2018.10.138

66.Yi JW, Zhang YC, Liao KC. Regional differential decomposition and formation mechanism of dynamic carbon emission efficiency of china's logistics industry. *Int J Environ Res Public Health.* (2021) 18:13121. doi: 10.3390/ijerph182413121

67. Hong HK, Liao HP, Li T, Yang J, Xie DT. Analysis of spatio-temporal patterns of rural space function based on entropy value method and Dagum Gini coefficient. *Trans Chin Soc Agric Eng.* (2016) 32:240–8. doi: 10.11975/j.issn.1002-6819.2016.10.034

68. Domma F, Condino F, Giordano S. A new formulation of the Dagum distribution in terms of income inequality and poverty measures. *Physica A*. (2018) 511:104–26. doi: 10.1016/j.physa.2018.07.027

69. Zambom AZ, Ronaldo D. A review of kernel density estimation with applications to econometrics. *Int Econom Rev.* (2013) 5:20–42. doi: 10.48550/arXiv.1212.2812

70. Horvath PA, Autry CW, Wilcox WE. Liquidity implications of reverse logistics for retailers: a Markov chain approach. J Retail. (2005) 81:191–203. doi: 10.1016/j.jretai.2005.07.003

71. Ou ZW, Li K, Jiang YHWHY, Gan WX. Research on emergency logistics guarantee mechanism. *Logist Technol.* (2005) 9:13–5. doi: 10.3969/j.issn.1005-152X.2005.09.005

72. Xi MH. Emergency logistics management. Beijing China: Emergency Management Press (2021).

73. Drabek TE. Managing the emergency response. *Public Adm Rev.* (1985) 45:85–92. doi: 10.2307/3135002

74. Henstra D. Evaluating local government emergency management programs: what framework should public managers adopt? *Public Adm Rev.* (2010) 70:236–46. doi: 10.1111/j.1540-6210.2010.02130.x

75. Kapucu N, Arslan T, Demiroz F. Collaborative emergency management and national emergency management network. *Disaster Prev Manag.* (2010) 19:452–68. doi: 10.1108/09653561011070376

76. Kapucu N. Disaster and emergency management systems in urban areas. *Cities*. (2012) 29:S41–9. doi: 10.1016/j.cities.2011.11.009

77. Glick J. Enhancing logistics efficiency through computer and network technologies: a pathway to industry Informatization. *Acad J Sci Technol.* (2024) 10:142–4. doi: 10.54097/e05gkp44

78. Fujita M, Krugman PR, Venables A. The spatial economy: cities, regions, and international trade. Cambridge, Massachusetts, United States: MIT Press (2001).

79. Myrdal Gunnar, Sitohang Paul. "Economic theory and under-developed regions." London: Duckworth (1957).